Author's response to reviews

Title: Development of an attention-touch control for manual cervical distraction: a pilot randomized clinical trial for patients with neck pain

Authors:

Maruti R Gudavalli (Gudavalli_r@palmer.edu)
Stacie A Salsbury (stacie.salsbury@palmer.edu)
Robert D Vining (robert.vining@palmer.edu)
Cynthia R Long (LONG_C@palmer.edu)
Lance Corber (CORBER_L@palmer.edu)
Avinash G Patwardhan (apatwar@lumc.edu)
Christine M Goertz (christine.goertz@palmer.edu)

Version: 7  Date: 19 May 2015

Author's response to reviews: see over
May 19, 2015

Trials
C/O BioMed Central
236 Gray’s Inn Road
London WC1X 8HB
United Kingdom

Dear Editors-in-Chief,

We thank the reviewers for their valuable comments on our manuscript, “Manual cervical distraction for neck pain: results of a pilot randomized controlled trial” (Trials manuscript ID# 7660216961494920). We revised the manuscript based on these comments, including modifying the title per the recommendations of Reviewer 3 to “Development of an attention and touch control for manual cervical distraction: a pilot randomized clinical trial for patients with neck pain”. A point by point response to the reviewers’ comments are uploaded along with the revised manuscript showing the changes in light blue highlight. We are grateful for the constructive comments of the reviewers which have enhanced the presentation of the study findings. We also thank the editor for giving us the opportunity to improve and edit the manuscript and we are satisfied with this submission.

We are looking forward to further consideration of the revised manuscript for publication in Trials.

Respectfully

Maruti Ram Gudavalli, PhD
Associate Professor and
Director of Biomechanics Core
Palmer Center for Chiropractic Research
741 Brady Street
Davenport, IA 52803
Tel. (563) 884-5260
Fax (563) 884-5227
Gudavalli_r@palmer.edu